Author's response to reviews

Title: The predictive value of mental health for long-term sickness absence: The Major Depression Inventory (MDI) and the Mental Health Inventory (MHI-5) compared

Authors:

Sannie V Thorsen (svt@nrcwe.dk)
Reiner Rugulies (rer@nrcwe.dk)
Pernille U Hjarsbech (pmi@nrcwe.dk)
Jakob B Bjorner (jbj@nrcwe.dk)

Version: 6 Date: 9 September 2013

Author's response to reviews:

Thank you for accepting our manuscript for publication.

I have corrected the typo.

regards Sannie Thorsen